Author’s response to reviews

Title: Management and Outcomes of Patients Presenting with Sepsis and Septic Shock to the Emergency Department during Nursing Handover: A Retrospective Cohort Study

Authors:

Sami Alsolamy (solamys@ngha.med.sa)
Atheer Alsabhan (atheer.sabhan@gmail.com)
Najla Alassim (alassmina@NGHA.MED.SA)
Musharaf Sadat (sadamutu@ngha.med.sa)
Eman Al Qasim (alqasimem@ngha.med.sa)
Hani Tamim (hani_t@hotmail.com)
Yaseen Arabi (yaseenarabi@yahoo.com)

Version: 2 Date: 09 Jan 2018

Author’s response to reviews:

09.01.2018
Guangde Tu
Biomed Central

Dear Dr Guangde Tu

We thank you for considering our manuscript, “EMMD-D-17-00058R1- Management and Outcomes of Patients Presenting with Sepsis and Septic Shock to the Emergency Department during Nursing Handover: A Retrospective Cohort Study” for publication in BMC Emergency Medicine.
We have addressed all the comments in the following reply and made changes to the manuscript, which has significantly improved and enriched the manuscript. We hope that the manuscript will be accepted for publication.

Thank you for your considerations.

Sincerely,

Comment 1

Please make sure that all declarations listed below are provided in your manuscript.

List of abbreviations

Declarations

- Ethics approval and consent to participate

- Consent to publish

- Availability of data and materials

- Competing interests

- Funding

- Authors' Contributions

- Acknowledgements
Comment 2

Please ensure to provide a separate 'Conclusions' section after the discussion. This should state clearly the main conclusions and provide an explanation of the importance and relevance of the study reported.

Reply

The “Conclusion” section has been provided as requested.

Comment 3

Please add a section ‘Figures legend’ (usually after the references) where you list the following information for each figure:

Figure name
Description of a figure

Reply

Done

Comment 4

On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy.

Reply

Done